Author's response to reviews

Title: Evaluation of Coronary Blood Flow during Cardiac Arrest with Circulation Maintained through Mechanical Chest Compressions in a Porcine Model

Authors:

Henrik Wagner (henrik.wagner@skane.se)
Bjarne Madsen Hardig (bjarne.madsenhardig@jolife.com)
Stig Steen (stig.steen@med.lu.se)
Trygve Sjoberg (trygve.sjoberg@med.lu.se)
Jan Harnek (jan.harnesk@skane.se)
Goran K Olivecrona (goran.olivecrona@med.lu.se)

Version: 8 Date: 11 December 2011

Author's response to reviews: see over
Dear Sir/Madam

Thank you for your acceptance of our manuscript original article entitled “Evaluation of Coronary Blood Flow Velocity during Cardiac Arrest with Circulation Maintained through Mechanical Chest Compressions in a Porcine Model”. (MS: 1563319414558372)

We wish to thank the editor for the excellent reviews performed during the submission process.

We have followed all the Major and minor revisions as outlined by the email from Dec 5, 2011 regarding author list, author contribution, author affiliation, author qualifications and author competing interests. We have removed the highlighting in the text and reformatted figures and tables according to recommendations. We have also made some corrections of smaller grammatical errors.

Please contact if there is anything else we can do.

Sincerely,

Goran Olivecrona, MD., PhD
Dept of Cardiology
Skane University Hospital
Lund University
221 85 Lund
SWEDEN

goran.olivecrona@med.lu.se